Author's response to reviews

Title: Body mass index and overweight in relation to residence distance and population density: experience from the Northern Finland Birth Cohort 1966

Authors:

Simo Näyhä (simonayha@oulu.fi)
Tiina Lankila (tiinalankila@oulu.fi)
Arja Rautio (arjarautio@oulu.fi)
Markku Koiranen (markku.koiranen@oulu.fi)
Tuija H Tammelin (tuijah.tammelin@likes.fi)
Anja Taanila (anja.taanila@oulu.fi)
Jarmo Rusanen (jarmorusanen@oulu.fi)
Jaana Laitinen (jaanalaaitinen@ttl.fi)

Version: 2  Date: 10 April 2013

Author's response to reviews:

BMC Public Health 10 April 2013
Ms Audrey Ann Reyes
MS: 1726313152961977
Body mass index and overweight in relation to residence distance and population density: experience from the Northern Finland Birth Cohort 1966 Simo Näyhä, Tiina Lankila, Arja Rautio, Markku Koiranen, Tuija H Tammelin, Anja Taanila, Jarmo Rusanen and Jaana Laitinen

Dear Ms Audrey Ann Reyes,

In Your email of 9 April You asked us to say more clearly that the above manuscript has been accepted by the Ethical Committee.

Now, the sentence (the last sentence on page 3):

"The study was reviewed by the Ethical Committee of the Northern Ostrobothnia Hospital District."

has been changed to:

"The study was approved by the Ethical Committee of the Northern Ostrobothnia Hospital District."

There are no other changes.

Sincerely,
Simo Näyhä  
Professor MD PhD  
Institute of Health Sciences  
PO Box 5000, FIN-90014 University of Oulu, Finland  
phone: +358 40 5646 159  
fax: +358 8 5375 661  
email: simo.nayha@oulu.fi